



FEDERAL RESERVE BANK OF CLEVELAND

Working papers of the Federal Reserve Bank of Cleveland are preliminary materials circulated to stimulate discussion and critical comment on research in progress. They may not have been subject to the formal editorial review accorded official Federal Reserve Bank of Cleveland publications. The views stated herein are those of the authors and are not necessarily those of the Federal Reserve Bank of Cleveland or of the Board of Governors of the Federal Reserve System.

Working papers are available at:

www.clevelandfed.org/research.

Nowcasting U.S. Headline and Core Inflation

Edward S. Knotek II and Saeed Zaman

Forecasting future inflation and nowcasting contemporaneous inflation are difficult. We propose a new and parsimonious model for nowcasting headline and core inflation in the U.S. consumer price index (CPI) and price index for personal consumption expenditures (PCE) that relies on relatively few variables. The model's nowcasting accuracy improves as information accumulates over a month or quarter, outperforming statistical benchmarks. In real-time comparisons, the model's headline inflation nowcasts substantially outperform those from the Blue Chip consensus and the Survey of Professional Forecasters. Across all four inflation measures, the model's nowcasting accuracy is comparable to the Federal Reserve Board's Greenbook.

Keywords: inflation, nowcasting, forecasting, professional forecasters, Greenbook, real-time data.

JEL classifications: E3, E37, C53

Suggested citation: Knotek, Edward S. II, and Saeed Zaman, 2015. "Nowcasting U.S. Headline and Core Inflation," Federal Reserve Bank of Cleveland, working paper no. 14-03R.

Edward S. Knotek II and Saeed Zaman are at the Federal Reserve Bank of Cleveland. They thank the editor Pok-sang Lam, two anonymous referees, Todd Clark, Randy Verbrugge, their discussant Michele Modugno, and participants at the Federal Reserve System Committee on Macroeconomics meeting, the International Institute of Forecasters' 34th International Symposium on Forecasting, the 8th International Conference on Computational and Financial Econometrics, and the Eastern Economics Meeting for very helpful comments, and Bill Bednar for help in compiling the real-time data set. They are grateful to Michele Modugno for sharing his dynamic factor model code, and for MIDAS code by Eric Ghysels.

*First posted in May 2014.

I. Introduction

Inflation developments are important to a wide swath of economic actors, and projections of future inflation influence the present behavior of financial market participants, consumers, firms, and central banks. Unfortunately, a long literature has documented that inflation is difficult to forecast accurately. These difficulties extend to contemporaneous forecasting, or nowcasting, of the inflation rate in the current month or quarter. As a result, the best available benchmarks for current quarter inflation nowcasting come from surveys of professional forecasters who employ a range of objective and subjective information. We present a relatively parsimonious statistical model that in many cases outperforms these benchmarks in inflation nowcasting accuracy.

Our model nowcasts U.S. headline and core consumer inflation as measured by the consumer price index (CPI) and the price index for personal consumption expenditures (PCE) using a judiciously chosen small number of data series at different frequencies. Within our model, high-frequency data affect monthly nowcasts, and monthly nowcasts aggregate to quarterly nowcasts. To take advantage of the sequencing of incoming data over a month or quarter, the model features time-varying weights on disaggregate and aggregate variables in forecasting the aggregate coupled with deterministic model switching that depends on the available information set; disaggregates are only used when sufficient data are available to make them informative. Beyond these time-varying weights, we follow the literature that has emphasized the benefits of simplicity in inflation forecasting—as notably embodied in Atkeson and Ohanian (2001), among others—and rely on univariate and simple multivariate techniques estimated over short rolling windows. These short rolling windows, along with high-frequency energy price data, play a key role in improving nowcasting accuracy. We view the parsimony of

2

the model as a virtue, given the difficulties in forecasting inflation and the risks of overfitting forecasting exercises to historical patterns that may not persist into the future.

Taking the model to the data requires real-time data. This is especially true for PCE inflation, but monthly and quarterly CPI inflation readings using seasonally adjusted data are also subject to substantive revisions: revisions to headline CPI inflation are as large as revisions to headline PCE in absolute terms, though core CPI inflation revisions are slightly smaller than those for core PCE. Unfortunately, the availability of real-time data limits tests of the model to a relatively short time span, with the earliest readings available for 1999.

Over this time, we show that the model's nowcasts outperform a variety of statistical benchmarks, including other models that are used for nowcasting. We then compare the model's performance with arguably the best available benchmarks (see, e.g., Faust and Wright 2013): subjective nowcasts from professional forecasters, both aggregates from private forecasters—as captured by the Blue Chip Economic Indicators consensus and the median forecast from the Federal Reserve Bank of Philadelphia's Survey of Professional Forecasters (SPF)---and those from the Federal Reserve Board of Governors staff in the Greenbook. Despite the fact that the comparison period includes volatile times that might be expected to favor subjective forecasts including large swings in world oil prices, a financial crisis, and a deep recession—the model's nowcasts in many cases outperform those from professional forecasters. In real-time out-ofsample comparisons, the model's nowcasts of headline CPI inflation outperform those from the Blue Chip consensus, with especially large outperformance as the quarter goes on. The model's nowcasts for headline CPI and PCE inflation also outperform those from the SPF, with similar nowcasting accuracy for core inflation measures. Across all four inflation measures, the model's nowcasting accuracy is comparable to that of the Greenbook.

3

Improving upon inflation nowcasting is not only of interest for its own sake. Faust and Wright (2013) and Del Negro and Schorfheide (2013) show that inflation forecasts at longer horizons benefit by employing more accurate conditioning via nowcasts. In another context, Branch (2014) utilizes inflation nowcasts to estimate Taylor rules. Thus, this paper's compact model has broad applications to academic economists and professional forecasters.

Our paper marks a departure from much of the nowcasting literature. In contrast to research that extracts common factors from a large number of data series, we judiciously choose a small number of data series at different frequencies to inform our nowcasts and do not use factor models. While the seminal nowcasting working paper of Giannone et al. (2006) originally considered both GDP and inflation, much of the nowcasting literature has proceeded to focus on GDP, following the published version of Giannone et al. (2008). In one exception, Modugno (2013) applies a dynamic factor model—with a larger number of monthly, weekly, and daily data series compared with the limited set of variables we work with-to nowcast year-over-year U.S. CPI inflation. Monteforte and Moretti (2013) use a combination of a dynamic factor model to construct a measure of core inflation and mixed frequency data in the context of a mixed data sampling (MIDAS) regression model based on Ghysels et al. (2004, 2005) to nowcast year-overyear euro area inflation. We present results for nowcasting U.S. monthly, year-over-year, and quarterly inflation, especially because the latter is the usual jumping-off point for economists doing quarterly forecasting exercises. Nevertheless, despite its different structure, our inflation nowcasting model shares the finding in the nowcasting literature that as time passes and additional information arrives, nowcasts of the current period become more accurate on average (e.g., Bańbura et al. 2013).

This paper proceeds as follows. Section II presents the inflation nowcasting model. Section III discusses the real-time data. Section IV assesses the model's performance for monthly, year-over-year, and quarterly nowcasting, and Section V compares the model's inflation nowcasting accuracy with other forecasters. Section VI analyzes the sensitivity of our model to alternative specifications, and Section VII concludes.

II. An Inflation Nowcasting Model

At its core, our model follows a parsimonious approach to nowcasting inflation. First, we rely on a judiciously chosen set of data series to inform our estimates. Second, we combine simple univariate and multivariate regression techniques. Third, we impose time-varying weights on disaggregate and aggregate variables in nowcasting the aggregate which deterministically depend on the available information set at a point in time, thereby taking advantage of the information flow to improve nowcasting accuracy. Disaggregate information is used for nowcasting the aggregate, in the spirit of Hendry and Hubrich (2011)—but only when this information is available and informative, resulting in time-varying weights, as discussed in Lütkepohl (2010).

Our modeling approach focuses on nowcasting or near-term forecasting monthly inflation rates. In the United States, monthly inflation is usually reported in non-annualized terms as $\pi_t = 100(P_t/P_{t-1}-1)$, where P_t is the price level in month t. Based on monthly price levels, we follow the usual conventions of statistical agencies to compute year-over-year inflation rates as $\pi_{t,t-12} = 100(P_t/P_{t-12}-1)$ or quarterly inflation rates π_T measured at seasonally adjusted annualized rates as $\pi_T = 100[(P_T/P_{T-1})^4 - 1]$, where P_T denotes the price level in quarter T, which

5

is the average of the three monthly price levels in that quarter: $P_T = (1/3)(P_{T,t=1} + P_{T,t=2} + P_{T,t=3})$.¹ We maintain consistency with this method of computing inflation: we keep track of available monthly price levels and then nowcast or forecast the missing monthly readings of a given quarter to construct quarterly inflation rates.

Our nowcasting model takes the general form

$$\mathbf{A}_{s(t)}\mathbf{Z}_{t} = \mathbf{B}_{s(t)} + \mathbf{C}_{s(t)}\mathbf{X}_{t} + \sum_{j=1}^{J} \mathbf{D}_{j,s(t)}\mathbf{Z}_{t-j} + \boldsymbol{\varepsilon}_{s(t)}, \qquad (1)$$

where \mathbf{Z}_t is an $n \times 1$ vector of aggregates, \mathbf{X}_t is an $m \times 1$ vector of disaggregates that are informative over \mathbf{Z}_t , and $\boldsymbol{\varepsilon}_{s(t)} \sim N(\mathbf{0}, \boldsymbol{\Sigma})$. The coefficient matrices \mathbf{A} , \mathbf{B} , \mathbf{C} , and \mathbf{D}_j are $n \times n$, $n \times 1$, $n \times m$, and $n \times n$, respectively, and are allowed to vary over time depending on the available information set, denoted s(t); in particular, \mathbf{C} and \mathbf{D}_j measure the weights put on the disaggregates and lagged aggregates, respectively.

This model structure allows us to incorporate information from diverse sources. First, given energy prices' role in headline price index volatility, high-frequency energy prices are a useful disaggregate to have in nowcasting headline inflation.² By contrast, when energy price volatility is tame, having core inflation as a disaggregate can be helpful given its large weight in headline inflation. Second, the timing of data releases affects nowcasts. The U.S. Bureau of Labor Statistics (BLS) releases the CPI around the middle of the following month; e.g., the May CPI is released around mid-June. An open question is the availability of higher frequency data with predictive content that would be available prior to the release of the CPI. The Bureau of Economic Analysis (BEA) typically releases the other major measure of consumer prices, the

¹ Notably, this formula is consistent with the way that the U.S. Bureau of Economic Analysis and the Blue Chip Economic Indicators survey report quarterly inflation rates.

² Modugno (2013) also discusses the importance of high-frequency energy prices in nowcasting inflation. Stock and Watson (2003) suggest predictive content from oil prices for U.S. inflation, but these can differ from the gasoline prices we use. In addition, Stock and Watson (2003) consider inflation forecasts over longer time horizons.

PCE price index, around the end of the following month; e.g., the May PCE price index is released around the end of June, after the CPI for May is released. While the contents and coverage of the two price indexes differ, the CPI has predictive content over the PCE price index during the interim period before the latter is released.

Nowcasting Core Inflation

High-frequency disaggregate data that have predictive content over core inflation are limited.³ There are similar limitations on the availability of real-time disaggregated core inflation series at the monthly frequency, such as core goods and core services series.⁴ Thus, if $\mathbf{Z}_{t} = [\pi_{t}^{\text{Core CPI}}, \pi_{t}^{\text{Core PCE}}]'$ is the aggregate of interest—where $\pi_{t}^{\text{Core CPI}}$ and $\pi_{t}^{\text{Core PCE}}$ are the monthly core CPI inflation rate and core PCE inflation rate in month *t*, respectively—then $\mathbf{X}_{t} = \mathbf{0}$ in equation (1).

Because CPI releases precede PCE releases, we take advantage of this timing mismatch. If we have monthly core CPI inflation through month t, $\pi_t^{\text{Core CPI}}$, but we only have core PCE inflation through month t-1, $\pi_{t-1}^{\text{Core PCE}}$, we bridge core CPI to core PCE to nowcast the as-yet-unreleased monthly core PCE inflation rate in month t. Conditional on being in this state, the time-varying weights in equation (1) become

³ We do not explore The Billion Prices Project at MIT as a potential disaggregate, but our framework could incorporate it in \mathbf{X}_{t} . While that data series may have predictive content for core inflation, it only began in 2008 and is based predominantly on goods prices from online retailers. By contrast, core price indexes place a large weight on services, and internet purchases and prices comprise a small share of spending on goods. The exchange rate could affect core inflation through its effects on import prices, a point we return to below in the context of a competing model.

⁴ Peach et al. (2013) find it useful for forecasting purposes to separate core CPI goods inflation and core CPI services inflation and model the series separately, but their forecasting horizon is four quarters. Unfortunately, sources such as ALFRED do not have a long time series of real-time monthly core goods inflation and monthly core services inflation, which prevents a similar disaggregation in this paper.

$$\mathbf{A}_{s(t)} = \begin{bmatrix} 0 & 0 \\ a_{21} & 1 \end{bmatrix}, \ \mathbf{B}_{s(t)} = \begin{bmatrix} 0 \\ b_2 \end{bmatrix}, \ \mathbf{D}_{j,s(t)} = \mathbf{0} \ \forall \ j \ .$$
(2)

The coefficients in equation (2) are estimated over a window of length τ to nowcast $\hat{\pi}_t^{\text{Core PCE}}$.

In all other cases, we rely on the spirit of Atkeson and Ohanian (2001), who find that inflation over the previous four quarters is a difficult forecasting benchmark to beat, and we forecast monthly core inflation $\hat{\mathbf{Z}}_t$ using recursive 12-month moving averages, by fixing

$$\mathbf{A}_{s(t)} = \mathbf{I}_{2}, \ \mathbf{B}_{s(t)} = \mathbf{0}, \ \mathbf{D}_{j,s(t)} = (1/12)\mathbf{I}_{2}, \ J = 12.$$
(3)

Thus, if we have data through time t-1 on $\mathbb{Z}_{t-1} = [\pi_{t-1}^{\text{Core PCE}}, \pi_{t-1}^{\text{Core PCE}}]'$, we use equation (3) to recursively generate forecasts for time $t, t+1, \ldots$. If we have data through time t on $\pi_{t}^{\text{Core CPI}}$ but only through time t-1 on $\pi_{t-1}^{\text{Core PCE}}$, we first use equation (2) to nowcast $\hat{\pi}_{t}^{\text{Core PCE}}$ and then use equation (3) to recursively generate forecasts for time $t+1, t+2, \ldots$, where $\hat{\pi}_{t}^{\text{Core PCE}}$ is included as an observation in taking the moving average. In this way, the arrival of a new core CPI reading affects its own forecast; because PCE release dates lag behind CPI, the arrival of core PCE inflation has no impact on core CPI. The arrival of a new core CPI reading also affects the nowcast for core PCE inflation for that month, and this nowcast in turn affects the core PCE inflation data come out for that month, the forecast for core PCE inflation is potentially affected again, and the process resets.

Nowcasting Headline Inflation

In addition to core prices, food prices and energy prices are other key disaggregates for headline inflation. In theory, high-frequency futures and spot market prices for raw food items could have predictive content over monthly consumer food inflation, denoted π_t^{Food} , and serve as useful disaggregate indicators \mathbf{X}_t for food inflation and thus headline inflation. However, raw food prices are a small determinant of retail food prices, and it is unclear which futures and spot market prices or price indexes would be most powerful in predicting food inflation. While we explore this possibility in the robustness section below, our baseline model follows the principle of parsimony and we forecast monthly food inflation as we did for monthly core inflation in the absence of disaggregate information: assuming we have data through month *t*-1, we forecast $\hat{\pi}_t^{\text{Food}} = (1/12) \sum_{j=1}^{12} \pi_{t-j}^{\text{Food}}$ and then recursively forecast $\hat{\pi}_{t+k}^{\text{Food}}$, k=1,2,...,5

Energy prices offer a contrast to food prices, because gasoline prices dominate fluctuations in consumer energy prices, and gasoline prices are heavily influenced by oil prices. Gasoline prices and oil prices are available at a higher frequency than monthly and can be used to nowcast gasoline price inflation after seasonal adjustment, denoted $\hat{\pi}_t^{\text{Gasoline}}$, which can also be used as one of the disaggregate variables in nowcasting headline inflation.

Underlying our nowcasts of gasoline price inflation is the fact that gasoline prices tend to revert toward the level predicted by the most recently observed oil price. We implement this

⁵ The CPI and PCE price index treat food differently, suggesting there could potentially be two separate disaggregate series. In the CPI, the food index encompasses both food at home and food away from home, and the core CPI excludes all food. In the PCE price index, food and beverages purchased for off-premises consumption are classified as nondurable goods and are excluded from the core PCE price index. Food services and accommodations are classified as services and are included in the core PCE price index. This change took effect with the BEA's 2009 comprehensive revisions. Given real-time data limitations discussed in more detail below, we consider a single CPI food series that is used as a disaggregate measure for both CPI and PCE inflation.

relationship as follows. Suppose that gasoline price data are not available within month *t* but are at least partially available in month *t*-1. Let $P_{t-1}^{\text{Gasoline (NSA)}}$ be the average of those available nonseasonally adjusted prices, and let $\pi_{t-1}^{\text{Gasoline (NSA)}}$ denote the associated gasoline price inflation in month *t*-1. Letting P_{t-1}^{Oil} denote the average of available oil price readings within month *t*-1, assume that oil prices follow a random walk at a daily frequency to extend the monthly oil price series by one additional monthly observation to \hat{P}_t^{Oil} .⁶ Because movements in oil prices pass through to gasoline prices, the length of the gasoline inflation series can be extended to month *t* to match the length of the oil price series following a two-stage regression. First, we posit a longer-run relationship between oil and gasoline prices:

$$P_{t}^{\text{Gasoline (NSA)}} = \alpha + \beta P_{t}^{\text{Oil}} + error_{1,t}; \qquad (4)$$

we let $\tilde{P}_t^{\text{Gasoline (NSA)}}$ denote the predicted gasoline price based on equation (4). Second, we posit an error correction model that incorporates the lagged discrepancy between gasoline prices and their predicted value:

$$\Delta P_{t}^{\text{Gasoline (NSA)}} = b \Delta P_{t}^{\text{Oil}} + c \left(P_{t-1}^{\text{Gasoline (NSA)}} - \tilde{P}_{t-1}^{\text{Gasoline (NSA)}} \right) + error_{2,t}.$$
(5)

Equations (4) and (5) can be estimated over a window τ_L of available data to capture these longer-run relationships.⁷ The estimated coefficients are combined with the oil price forecast \hat{P}_t^{Oil} to produce forecasts $\hat{P}_t^{\text{Gasoline (NSA)}}$ and $\hat{\pi}_t^{\text{Gasoline (NSA)}}$. Finally, because there is a seasonal pattern in gasoline prices that is removed from monthly (seasonally adjusted) inflation figures,

⁶ Omitting oil futures prices further limits the number of variables needed without sacrificing forecasting accuracy; see Alquist and Kilian (2010) for evidence that a no-change (random walk) forecast can beat futures prices as a near-term predictor of oil prices. When nowcasting quarterly inflation rates and hence multiple monthly inflation nowcasts are required, we still extend oil prices by one monthly observation; if we have P_t^{Oil} , we use the last daily

oil price observation to produce $\hat{P}_{t+1}^{\text{Oil}}$.

⁷ Imposing $b = \hat{\beta}$, equation (5) takes the form $\left(P_{t}^{\text{Gasoline (NSA)}} - \tilde{P}_{t}^{\text{Gasoline (NSA)}}\right) = a\left(P_{t-1}^{\text{Gasoline (NSA)}} - \tilde{P}_{t-1}^{\text{Gasoline (NSA)}}\right) + error_{2,t}$.

we seasonally adjust the gasoline price inflation forecast for month *t*, $\hat{\pi}_t^{\text{Gasoline (NSA)}}$, to produce $\hat{\pi}_t^{\text{Gasoline}}$.⁸ Given release lags, we typically have one or two more months of gasoline inflation nowcasts or forecasts, $\hat{\pi}_{t+k}^{\text{Gasoline}}$, $k \ge 0$, than we have inflation data on the other series.

If gasoline price data are available for all or part of month *t*, then we let $P_t^{\text{Gasoline (NSA)}}$ be the average of the available non-seasonally adjusted prices, we use them to compute monthly gasoline inflation, $\pi_t^{\text{Gasoline (NSA)}}$, and we seasonally adjust the data to nowcast $\hat{\pi}_t^{\text{Gasoline}}$, which is a nowcast because we may not have all of the gasoline price data for the month and because we do not have the exact seasonal factor. As additional high-frequency gasoline price data arrive during month *t*, the values of $\pi_t^{\text{Gasoline (NSA)}}$ and $\hat{\pi}_t^{\text{Gasoline}}$ are updated.

Finally, we can construct nowcasts and forecasts of headline inflation rates using the model in equation (1) and weights that vary deterministically with the available state of information. Let $\mathbf{Z}_t = [\pi_t^{\text{CPI}}, \pi_t^{\text{PCE}}]'$ be the aggregate of interest, where π_t^{CPI} and π_t^{PCE} are the monthly CPI inflation rate and PCE inflation rate in month *t*, respectively. The vector of relevant disaggregates for headline inflation is

$$\mathbf{X}_{t} = \left[\pi_{t}^{\text{Core CPI}}, \pi_{t}^{\text{Core PCE}}, \pi_{t}^{\text{Food}}, \pi_{t}^{\text{Gasoline}}\right]'.$$
(6)

For states in which we have π_t^{CPI} but not π_t^{PCE} , we again bridge the headline CPI reading to headline PCE; the time-varying weights

⁸ We seasonally adjust our measure of gasoline price inflation as follows. Let $\pi_{t,Y}^{\text{CPI, Gasoline}}$ denote monthly inflation in the seasonally adjusted CPI for gasoline in month *t* of year *Y*, and let $\pi_{t,Y}^{\text{Gasoline}(NSA)}$ denote our measure of monthly gasoline price inflation based on high-frequency data in month *t* of year *Y*. We construct a seasonal factor for month *t* in year *Y* by taking the average difference between the non-seasonally adjusted and seasonally adjusted measures for the same month in the preceding three years: $s_{t,Y} = (1/3) \sum_{j=1 \text{ year}}^{3 \text{ years}} (\pi_{t,Y-j}^{\text{Gasoline}(NSA)} - \pi_{t,Y-j}^{\text{CPI, Gasoline}})$. Because it is based on information from previous years, we can apply this seasonal factor to derive a seasonally adjusted nowcast of gasoline price inflation in month *t* of year *Y*: $\hat{\pi}_{t,Y}^{\text{Gasoline}(NSA)} - s_{t,Y}^{f}$. Note that gasoline CPI readings only enter into seasonal adjustment; $\hat{\pi}_{t}^{\text{Gasoline}}$, where we suppress the year subscript, is our disaggregate of interest.

$$\mathbf{A}_{s(t)} = \begin{bmatrix} 0 & 0 \\ a_{21} & 1 \end{bmatrix}, \ \mathbf{B}_{s(t)} = \begin{bmatrix} 0 \\ b_2 \end{bmatrix}, \ \mathbf{C}_{s(t)} = \mathbf{0}, \ \mathbf{D}_{j,s(t)} = \mathbf{0} \text{ for all } j$$
(7)

can be estimated over a window of data of length τ to produce a nowcast $\hat{\pi}_{t}^{\text{PCE}}$. For states in which we have $\hat{\pi}_{t}^{\text{Gasoline}}$, we pair that nowcast with the forecasts of $\hat{\pi}_{t}^{\text{Food}}$, $\hat{\pi}_{t}^{\text{Core CPI}}$, and $\hat{\pi}_{t}^{\text{Core PCE}}$ generated earlier to complete the vector $\hat{\mathbf{X}}_{t}$. The time-varying weights

$$\mathbf{A}_{s(t)} = \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix}, \ \mathbf{B}_{s(t)} = \begin{bmatrix} b_1 \\ b_2 \end{bmatrix}, \ \mathbf{C}_{s(t)} = \begin{bmatrix} c_{11} & 0 & c_{13} & c_{14} \\ 0 & c_{22} & c_{23} & c_{24} \end{bmatrix}, \ \mathbf{D}_{j,s(t)} = \mathbf{0} \text{ for all } j$$
(8)

are estimated over a window of data of length τ and we can then forecast $\hat{\mathbf{Z}}_{t} = [\hat{\pi}_{t}^{\text{CPI}}, \hat{\pi}_{t}^{\text{PCE}}]'$ using $\hat{\mathbf{X}}_{t}$. Incoming high-frequency data on gasoline prices that affect $\hat{\pi}_{t}^{\text{Gasoline}}$ will affect $\hat{\mathbf{X}}_{t}$ and headline inflation nowcasts through equation (8), as will incoming data that affect the forecasts for food and core inflation. And in states for which we lack $\hat{\pi}_{t}^{\text{Gasoline}}$ and thus do not have the complete disaggregate vector $\hat{\mathbf{X}}_{t}$, we use recursive 12-month moving averages by fixing

$$\mathbf{A}_{s(t)} = \mathbf{I}_{2}, \ \mathbf{B}_{s(t)} = \mathbf{0}, \ \mathbf{C}_{s(t)} = \mathbf{0}, \ \mathbf{D}_{j,s(t)} = (1/12)\mathbf{I}_{2}, \ J = 12.$$
(9)

The use of past headline inflation (rather than core inflation) to predict future headline inflation is consistent with the results in Crone et al. (2013).

As with the procedure for core inflation set out earlier, nowcasts or forecasts of headline inflation can enter the recursion in equation (9) if they are part of the 12-month window. Also as with core inflation, because PCE inflation is released after CPI inflation, we only bridge from CPI to PCE if there is an additional CPI reading. Hence, nowcasts or forecasts for headline CPI inflation are determined by equations (8) or (9), whereas nowcasts or forecasts of headline PCE inflation are determined by equations (7), (8), or (9) depending on the available information set.

III. Data Sources and the Need for Real-Time Data

Implementing the model requires a number of monthly inflation series from the CPI and the PCE price index, along with higher frequency data on gasoline and oil prices. While long historical series are readily available from the BLS, the BEA, and data collection sites, both the CPI and the PCE price index are subject to data revisions—from new estimates of seasonal patterns in the case of the CPI to regular comprehensive revisions in the case of the PCE price index.⁹ Thus, the currently available historical time series may differ substantially from what would have been available to forecasters at some point in the past.

To assess the need for real-time data—which are generally more difficult to come by at the model's monthly frequency—we document the extent to which the most recent ("final") vintage data differ from real-time data using headline and core inflation in the CPI and PCE price index.¹⁰ Despite the fact that CPI measures are only subject to seasonal revisions and not the comprehensive revisions of the PCE measures, data revisions appear substantial when looking at either CPI or PCE inflation.¹¹ In the quarterly data, Figure 1(a) plots differences between initial

⁹ The non-seasonally adjusted CPI is not subject to revisions and is final when published, making year-over-year inflation rates computed from that index invariant to the passage of time. For monthly and quarterly inflation readings, however, the NSA CPI data are of little use because of predictable seasonal fluctuations.

¹⁰ Monthly real-time data come from the St. Louis Fed's Archival Federal Reserve Economic Data (ALFRED). The "final" vintage data were downloaded on August 19, 2015. Because the CPI went through July 2015 while the PCE price index went through June 2015, we treated the June 2015 CPI and PCE inflation readings as the last available observations. The initial inflation readings are the first available ones at the monthly or quarterly frequency. The comparisons begin in the middle of 2000, which correspond to the earliest availability of monthly real-time PCE inflation measures. We stop the comparison at the end of 2014 because the 2015 CPI readings have not been subject to revision yet. The CPI's seasonal factors for the previous five years are subject to revision once each year is complete: for example, after the December 2013 CPI was released, the BLS revised the seasonal factors for 2009 through 2013.

¹¹ Faust and Wright (2013) find that "revisions to CPI and core CPI inflation are trivial; but revisions to the other inflation measures are large" (p. 9). This may reflect the manner in which they compute revisions. Instead of using the final (most recent) vintage of data, they examine inflation as of the real-time rate recorded two quarters after the

and final vintage headline inflation, while Figure 1(b) plots the differences in core inflation. The revisions are large and do not appear markedly different for the headline inflation measures. Table 1 presents statistics on the differences for both quarterly inflation rates and monthly inflation rates. While the average revision is near zero, the average absolute revision was not negligible. Headline CPI revisions were larger in absolute terms and more volatile than headline PCE revisions during this time, while core CPI revisions were smaller than core PCE revisions.



Notes: Measures are initial readings less final readings. Final vintage inflation data were downloaded on August 19, 2015. To treat our inflation measures similarly, we consider the June 2015 CPI and PCE inflation observations to be the last available readings. The initial quarterly inflation reading is computed as soon as all of the monthly price index readings for the quarter are available.

					0			
	<u>C</u>	PI	Core	CPI	<u>PC</u>	<u>E</u>	Core	PCE
		Absolute		Absolute		Absolute		Absolute
	Difference	difference	Difference	difference	Difference	difference	Difference	difference
Monthly Data								
Average	0.00	0.08	0.00	0.03	-0.01	0.07	-0.02	0.05
Standard deviation	0.10	0.06	0.04	0.03	0.09	0.06	0.07	0.05
Quarterly Data								
Average	-0.03	0.51	0.00	0.21	-0.07	0.43	-0.14	0.33
Standard deviation	0.64	0.37	0.27	0.16	0.55	0.35	0.42	0.29
Matan Manthles inf	lation notes a		1:	4 - 1	la la mananta d	inflation		and in a d

Table 1: Differences between Initial and Final Vintage Inflation Rates

Notes: Monthly inflation rates are non-annualized percent changes, while quarterly inflation rates are annualized percent changes. Difference measures are initial readings less final readings. Final vintage inflation data were downloaded on August 19, 2015. To treat our inflation measures similarly, we consider the June 2015 CPI and PCE inflation observations to be the last available readings. The comparisons begin in June 2000 (2000Q2) and end in December 2014 (2014Q4).

quarter in question. Because the BLS revises the CPI only once per year, this methodology reduces the number of possible CPI revisions.

Because of these large revisions, using final vintage data and conducting pseudo realtime analysis would be problematic. This is especially true when comparing nowcasts to those from other forecasts made in real time by professional forecasters, as different information sets would contaminate the comparisons. As such, our analysis only utilizes series for which realtime historical data are available.

The St. Louis Fed's ALFRED database contains real-time vintages for the monthly PCE price index and core PCE price index starting with the June 2000 readings. Real-time monthly headline CPI coverage begins with June 1972, and monthly core CPI coverage begins with November 1996. The model also requires a measure of food inflation, and ALFRED has the real-time food CPI starting with November 1996.¹²

Higher frequency data are available for energy prices. Every Monday, the Energy Information Administration (EIA) publishes average retail gasoline prices for all grades based on a survey of approximately 800 retail gasoline outlets, with the series beginning in 1993. For oil prices, we use Brent crude spot prices from the Financial Times, which are available starting in 1987. We consider the robustness of our results to using West Texas Intermediate crude spot prices below. The final needed series is the seasonally adjusted CPI for gasoline, which is used to seasonally adjust retail gasoline prices from EIA. Combining data from ALFRED and Haver Analytics, we have real-time coverage of the gasoline CPI back to January 1999.

Thus, we can perform real-time out-of-sample nowcasting starting with February 1999 for the CPI and July 2000 for the PCE price index. In total, our nowcasting model uses only 8 data series—monthly CPI, core CPI, food CPI, gasoline CPI, PCE, and core PCE; weekly retail

¹² In the PCE price index, food services are part of core inflation, while food and beverages purchased for offpremises consumption are not. Having a measure of inflation in the latter could benefit headline PCE inflation nowcasts (as would the CPI equivalent, which is the food at home series), but these series are not available with long real-time histories. Thus, we rely on the CPI for food.

gasoline prices; and daily oil prices—though we have many data vintages to conduct the realtime analysis. Rather than incorporate components' weights explicitly, the model estimates the historical contributions of disaggregated series to the aggregate, including "other" effects in key unmodeled inflation components that are subsumed in the constant terms and may vary over time. Coupling these considerations with the need to estimate few model parameters, we use short rolling windows (τ =24 monthly observations) to capture potential time-variation in the coefficients. In order to ensure that our two-stage regression captures the longer-term relationship between oil and gasoline prices, we estimate it using a longer rolling window (τ_L =60 monthly observations). We consider robustness to rolling window sizes in Section VI.

We assess the nowcasting accuracy of our model along several dimensions. As one standard metric for point forecast evaluation, we examine root mean squared errors (RMSEs). The combination of real-time data, differing estimation schemes, and generated regressors in our model imposes challenges in evaluating the statistical significance in reductions in RMSEs from our model with plausible alternatives; while we report results from tests of conditional predictive ability based on Giacomini and White (2006), we view these as approximations.¹³ As a second metric, we assess nowcasting accuracy via the directional forecast approach of Pesaran and Timmermann (2009), as in Baumeister et al. (2015). Directional accuracy is measured by success ratios, defined as the percentage of times that the nowcasting model correctly predicted the change in the rate of inflation from the previous period; success ratios greater than 0.5 indicate improvement over a no-change forecast.

¹³ Clark and McCracken (2009) discuss issues with real-time data and tests of equal forecast accuracy. We report the Giacomini-White test based on our use of rolling windows in our baseline model, but several of the alternative statistical specifications we present use expanding windows. In our cases, the Giacomini-White test results are very similar to those from Diebold and Mariano (1995) tests for equal forecast accuracy, with the adjustment for small samples of Harvey et al. (1997) as applied to nowcasting in Carriero et al. (2015).

IV. Model Performance for Monthly, Year-Over-Year, and Quarterly Inflation

A common finding in the nowcasting literature is that as time passes and additional information arrives, nowcasts of the current period generally become more accurate on average (e.g., Bańbura et al. 2013). Our inflation nowcasting model shares this property, whether examining the ability of the model to nowcast monthly inflation, year-over-year inflation, or quarterly inflation. While we focus attention on a limited number of cases in this section, the model can produce nowcasts at a daily frequency.

An open question when evaluating real-time forecast accuracy is the choice of what constitutes the "actual" data realizations (or "truth"). Ex post revisions take a variety of forms, including new seasonal factors, the incorporation of more complete source data, and new methodologies. It is also difficult to know whether professional forecasters aim to forecast the initial data release, which is one measure of forecasting prowess, or whether their forecasts aim to capture subsequent revisions as well, which may or may not be mean zero in expectation. To incorporate more complete source data but not necessarily methodological revisions that may have been impossible to predict, we treat the third monthly estimate of PCE prices as "truth," similar to Tulip (2009) and a number of other researchers; we treat CPI symmetrically.^{14,15}

¹⁴ The third estimate was previously called the "first final" estimate; see Tulip (2009). At the end of the sample, we treat the last available reading as "truth." Note that Tulip (2009) uses quarterly data from the Philadelphia Fed's real-time database, so including the third estimate required using the real-time reading available two quarters later. ¹⁵ In computing all nowcasting accuracy statistics, we exclude nowcasts for PCE and core PCE inflation for the months of September 2001 and October 2001 in the monthly exercises, and 2001Q3 and 2001Q4 in the quarterly exercises, because these observations are extreme outliers in our short sample. The September 11, 2001, terrorist attacks triggered insurance payments that caused a large one-time drop in the PCE price index for insurance. The decline in this component was so large that monthly core PCE inflation for September 2001 fell to its lowest recorded reading. The decline was unwound in October 2001. CPI was not affected by these insurance payments.

Monthly inflation readings come out around the middle (CPI) or end (PCE price index) of the following month. Over the course of a given month t, the arrival of the month t-1 inflation estimate contains relevant information and influences the nowcast for the current month t. Oil prices and retail gasoline prices arrive at the daily and weekly frequency, respectively, and the flow of these data sources also impacts the nowcast.

We illustrate the model's monthly nowcasting performance for CPI and PCE inflation at six representative dates listed in Table 2. Assuming that month *t* is the target month being nowcasted, Case 1 is the final day of month t-1, and case 5 is the last day of the target month *t*. Case 6 is the middle of month t+1, when the CPI is released for month *t* and only the PCE price index is left to be nowcasted (in this case, backcasted) for month *t*.

	Date	Example: Nowcasting target month is January					
Case 1	Last day of the provious month	Last day of December, assume have CPI and PCE through					
	Last day of the previous month	November.					
Case 2	Day 9 of the target month	Have at least one weekly retail gasoline reading, have CPI and PCE					
	Day 8 of the target month	through November.					
Case 3	Day 15 of the target month	Have at least two weekly retail gasoline readings, assume receive					
	Day 15 of the target month	CPI for December, have PCE through November.					
Casa 4	Day 22 of the target month	Have at least three weekly retail gasoline readings, CPI through					
Case 4	Day 22 of the target month	December, PCE through November.					
Casa 5	Last day of the target month	Have all weekly retail gasoline readings, CPI through December,					
Case 5	Last day of the target month	assume receive PCE for December.					
0 (Dow 15 of the following month	Have all weekly retail gasoline readings, assume receive CPI for					
Case o	Day 15 of the following month	January, have PCE through December.					

Table 2: Monthly Nowcasting Performance Cases

We use the real-time data that would have been available in each assumed case to nowcast monthly inflation rates, running from September 2000 to June 2015. Figure 2 plots the monthly RMSEs from our baseline model; we also show the RMSEs in the table below.

The monthly core inflation RMSEs change little over time, consistent with using

recursive 12-month moving averages to forecast missing monthly data in our baseline model, but

they do drift progressively lower. Assuming month *t* is the target month to nowcast, the arrival of the core CPI reading for month t-1 (case 3) reduces the RMSE for core CPI by 0.002 percentage point. This also generates a reduction in RMSE for core PCE: the month t-1 core CPI inflation is bridged to nowcast month t-1 core PCE inflation, which is now one of the 12 observations used to nowcast the core PCE inflation rate in month *t*. There is a small reduction in RMSE upon the arrival of the month t-1 core PCE release (case 5), because this data point—instead of a nowcast—is now one of the 12 observations used to nowcast the core CPI for the target month *t* is released (case 6), bridging that reading to core PCE reduces RMSE by 0.021 percentage point. In a sense, this is the first available data release for the month *t* being nowcasted in terms of core inflation.





Notes: Case 1 is right before the start of the month. Case 2 is day 8 of the month. Case 3 is day 15 of the month, at which point the previous month's CPI is assumed to be available. Case 4 is day 22 of the month. Case 5 is the last day of the month, at which point the previous month's PCE price index is assumed to be available. Case 6 is day 15 of the following month, at which point the CPI for the month being forecasted is assumed to be available. Inflation rates are month-over-month percent changes, so errors are expressed in non-annualized percentage points. PCE and core PCE statistics exclude September and October 2001. The exercise uses real-time data from September 2000 through June 2015.

The pattern is different for headline inflation because of the availability of higher-

frequency energy prices within month t. Headline inflation RMSEs decline steadily and

significantly over the course of time. By day 8 of month *t* (case 2), when at least one weekly reading on retail gasoline prices is available for the month, RMSEs fall sharply from where they were immediately prior to the start of the month, and they move lower as additional gasoline price data accumulate. The accuracy of nowcasting headline PCE inflation benefits from the arrival of the monthly CPI readings, both in case 3 when the time *t*-1 CPI reading becomes available and in case 6 when the time *t* CPI reading becomes available. Immediately prior to the inflation releases for the targeted month *t*, nowcasting RMSEs for headline CPI and PCE are less than half of their values compared with where they stood on the final day of month *t*-1.¹⁶

We compare the model's monthly nowcasting performance with three competing models. The first alternative uses a random walk in monthly inflation, $\hat{\pi}_t = \pi_{t-k}$, where the nowcast for the target month *t* is based on the most recent available real-time monthly inflation rate from *k* months ago.

The second alternative is a mixed data sampling (MIDAS) model based on Ghysels et al. (2004, 2005). Monteforte and Moretti (2013) apply a MIDAS model to nowcast euro area inflation.¹⁷ For nowcasting or forecasting monthly inflation at horizon h, π_{t+h} , we consider a MIDAS with leads specification of the form:

$$\pi_{t+h} = \alpha_{(h)} + \sum_{j=0}^{P(M)-1} \chi_{j+1,(h)} \pi_{t-j} + \sum_{j=0}^{P(M)-1} \gamma_{j+1,(h)} Z_{t-j} + \beta_{(h)} \sum_{j=0}^{P(HF)-1} \omega_{P(HF)-j}(\theta_{(h)}^{HF}) X_{P(HF)-j,t+1}^{HF} + e_{t+h} .$$
(10)

For exposition, we assume that lagged dependent monthly variables are available through month t; other monthly variable(s) Z are also available through month t; P(M) is the number of lags of

¹⁶ Because of the very small number of data series in our model, we do not pursue a decomposition of nowcast revisions based on news shocks as in, e.g., Modugno (2013) or Bańbura et al. (2013), though such a decomposition could be presented in practice. On days in which neither the CPI nor the PCE price index is released (or revised), core inflation nowcasts are not revised and any revisions to headline inflation are due to energy (gasoline and oil) price movements. The deterministic model switching we implement would modestly complicate the interpretation of news shock; see Section VI for an alternative specification that omits model switching.

¹⁷ Monteforte and Moretti (2013) also use a factor model to generate a measure for core inflation, but we omit this step as we use core inflation measures from U.S. statistical agencies instead.

the monthly variables, which we set to one; at any given point in time, we have P(HF) highfrequency observations, $X_{1,t+1}^{HF}$,..., $X_{P(HF),t+1}^{HF}$, in month t+1, and we use all P(HF) of them as highfrequency leads. Our coefficients can vary with the forecast horizon, as captured by the (h)subscripts. As discussed in Ghysels (2015), we identify $\beta_{(h)}$ by assuming

 $\sum_{j=0}^{P(IF)-1} \omega_{P(HF)-j}(\theta_{(h)}^{HF}) = 1.$ We estimate the model via nonlinear least squares and parameterize the MIDAS polynomial with the Beta density. While Monteforte and Moretti (2013) construct three MIDAS models each with three daily data series geared toward nowcasting euro area inflation, we focus on parsimonious specifications using only the data series in our baseline model. Preliminary analysis favored this parsimonious approach over MIDAS regression models featuring a larger number of high-frequency indicators along the lines of Monteforte and Moretti (2013).¹⁸ To take advantage of potential gains coming from model averaging, some of which are documented in Andreou et al. (2013), we run two separate MIDAS regressions using our two high-frequency data series—the first with daily oil prices and a second with weekly gasoline prices, both of which enter the model in natural log first differences—and then construct an average nowcast or forecast for the monthly inflation series at horizon *h*.

The third alternative is a dynamic factor model (DFM) based on Modugno (2013) for nowcasting the U.S. CPI, and which built on earlier work by Giannone et al. (2006, 2008). Following Modugno (2013), the model combines data at the monthly, weekly, and daily

¹⁸ We tailor each set of explanatory variables *Z* to the inflation measure being nowcasted or forecasted and the variables available when making that nowcast or forecast. Doing so gives the MIDAS models the benefit of the deterministic model switching we propose in the baseline model. The dependent variable's own lag is always included. For core CPI inflation, *Z* only contains lagged CPI inflation. For CPI inflation, *Z* contains lagged core CPI inflation, lagged gasoline inflation, and lagged food inflation. For core PCE inflation, if there is one more core CPI reading than core PCE, then *Z* only contains contemporaneous core CPI inflation, similar to our baseline approach; otherwise, *Z* contains lagged core CPI inflation. For PCE inflation, we use the same convention: if there is one more CPI reading than PCE, then *Z* only contains contemporaneous CPI inflation; otherwise, *Z* contains lagged core PCE inflation, lagged gasoline inflation, and lagged food inflation.

frequencies into a business day frequency factor model with missing observations that are cast in a state space representation. The dynamic factor model takes the general form:

$$y_t = Cf_t + \varepsilon_t, \ \varepsilon_t \sim N(0, \Sigma) \tag{11}$$

with y_t a vector of observations, *C* a matrix of loadings, ε_t a vector of idiosyncratic components, and f_t a vector of unobserved common components following a VAR given by

$$Bf_{t} = A(L)f_{t-1} + u_{t}, \ u_{t} \sim N(0,Q),$$
(12)

where *B* and A(L) are matrices governing factor dynamics, some of which may be time-varying. Modugno (2013) shows how monthly, weekly, and daily variables and factors can be stacked appropriately in equations (11) and (12) and how the unobserved daily factors aggregate to weekly and monthly factors, which in turn inform nowcasts and forecasts of the monthly variables. As in Modugno (2013), we estimate relevant parameters via the approach of Bańbura and Modugno (2012). Our dynamic factor model dataset is slightly smaller than in Modugno (2013), but our nowcasting results are highly comparable.¹⁹ We report results for a model specification with 1 factor and 6 lags, which we found generated the most accurate out-of-sample nowcasts in the specifications we considered.

Table 3 shows that our nowcasting model has historically generated lower RMSEs than the three competing model nowcasts. The MIDAS models and the dynamic factor model generated lower RMSEs in only one case each. In many cases for headline CPI and PCE inflation, the reductions in RMSEs coming from the baseline model compared with the

¹⁹ Modugno (2013) used 8 monthly, 4 weekly, and 15 daily series. Due to data availability and real-time data limitations, along with our interest in nowcasting headline and core inflation for the CPI and PCE price index, we use 6 monthly variables (CPI inflation, core CPI inflation, PCE inflation, core PCE inflation, food inflation, gasoline inflation), 4 weekly variables (diesel fuel price, regular grade retail gasoline price, midgrade retail gasoline price, and premium retail gasoline price), and 14 daily series (Brent crude oil, foodstuffs price index, grains price index, fats and oils price index, raw sugar price, raw industrials price index, agricultural commodities price index, textiles and fibers price index, industrial metals price index, steel scrap prices, the 10-year Treasury note constant maturity yield, the 3-month Treasury bill rate, the S&P 500 stock price index, and the nominal trade-weighted exchange value of the dollar against major currencies).

alternative models are substantial. In the case of core inflation, our baseline model is highly competitive with the alternative approaches in spite of its extreme parsimony; there is little benefit from the more sophisticated and computationally intensive MIDAS models and dynamic factor model, even though the latter includes some high-frequency information that could have a bearing on core inflation (e.g., the exchange rate).

		Case					
Measure	Model	1	2	3	4	5	6
CPI	Baseline	0.294	0.194	0.159	0.139	0.133	
	Random walk	0.480***	0.480***	0.366***	0.366***	0.366***	
	MIDAS	0.303	0.296***	0.258***	0.261***	0.250***	
	DFM	0.305	0.265***	0.264***	0.268***	0.273***	
Core CPI	Baseline	0.089	0.089	0.087	0.087	0.087	
	Random walk	0.112***	0.112***	0.106***	0.106***	0.106***	
	MIDAS	0.091	0.090	0.090	0.090	0.091	
	DFM	0.090	0.091	0.091	0.091	0.091	
PCE	Baseline	0.205	0.139	0.117	0.104	0.097	0.070
	Random walk	0.344*	0.344***	0.344***	0.344***	0.263***	0.263***
	MIDAS	0.244*	0.250***	0.240***	0.240***	0.171***	0.056**
	DFM	0.212	0.186***	0.187***	0.188***	0.189***	0.186***
Core PCE	Baseline	0.078	0.078	0.077	0.077	0.075	0.055
	Random walk	0.159*	0.159*	0.159*	0.159*	0.120	0.120*
	MIDAS	0.104	0.104	0.082	0.081	0.096	0.059*
	DFM	0.078	0.080	0.082	0.084**	0.082**	0.081***

 Table 3: Root Mean Squared Nowcast Errors, Monthly Inflation

Notes: Case 1 is right before the start of the month. Case 2 is day 8 of the month. Case 3 is day 15 of the month, at which point the previous month's CPI is assumed to be available. Case 4 is day 22 of the month. Case 5 is the last day of the month, at which point the previous month's PCE price index is assumed to be available. Case 6 is day 15 of the following month, at which point the CPI for the month being nowcasted is assumed to be available. *, **, and *** denote rejection of the null of equal conditional predictive ability for the baseline model compared with each alternative model at the 10%, 5%, and 1% level, respectively, based on the Giacomini-White test. **Bold** entries denote that the baseline model produces smaller RMSEs than the alternative model. Inflation rates are month-overmonth percent changes in seasonally adjusted data, so errors are expressed in non-annualized percentage points. PCE and core PCE statistics exclude September and October 2001. The exercise uses real-time data from September 2000 through June 2015.

The top panel of Table 4 shows the success ratios for our baseline model and the two competing nowcasting models, computed as the percentage of nowcasts that correctly predicted whether inflation increased or decreased compared with the preceding monthly reading. The bottom panel shows the *p*-values from the Pesaran-Timmermann test for directional forecast

accuracy. For headline inflation, our baseline model generates considerably higher success ratios and lower *p*-values almost across-the-board. For core inflation, meanwhile, the results are mixed, suggesting that our parsimonious baseline model is highly competitive with MIDAS models and dynamic factor models.²⁰

				Ca	se		
Measure	Model	1	2	3	4	5	6
			Success rat	ios			
CPI	Baseline	0.753	0.815	0.876	0.921	0.916	
	MIDAS	0.719	0.663	0.798	0.803	0.826	
	DFM	0.449	0.539	0.815	0.798	0.815	
Core CPI	Baseline	0.528	0.528	0.652	0.652	0.652	
	MIDAS	0.584	0.579	0.652	0.657	0.652	
	DFM	0.562	0.517	0.702	0.697	0.708	
PCE	Baseline	0.697	0.781	0.876	0.899	0.876	0.916
	MIDAS	0.618	0.562	0.472	0.478	0.758	0.927
	DFM	0.472	0.506	0.556	0.590	0.764	0.775
Core PCE	Baseline	0.500	0.500	0.624	0.624	0.719	0.798
	MIDAS	0.539	0.494	0.539	0.506	0.713	0.792
	DFM	0.539	0.506	0.489	0.522	0.713	0.713
		Pesard	an-Timmerma	nn p-values			
CPI	Baseline	2.4E-10	1.1E-16	0	0	0	
	MIDAS	7.2E-9	4.4E-6	4.4E-13	1.4E-13	1.1E-16	
	DFM	0.257	0.181	3.4E-15	8.3E-13	4.6E-14	
Core CPI	Baseline	0.522	0.522	0.011	0.011	0.011	
	MIDAS	0.046	0.059	1.7E-3	9.9E-4	1.8E-4	
	DFM	0.149	0.431	5.4E-6	1.7E-5	1.6E-6	
PCE	Baseline	4.6E-7	3.9E-14	0	0	0	0
	MIDAS	6.8E-4	0.014	0.215	0.326	7.1E-10	0
	DFM	0.807	0.422	0.017	1.8E-04	6.7E-9	4.2E-10
Core PCE	Baseline	0.954	0.954	0.025	0.025	4.4E-7	1.1E-14
	MIDAS	0.405	0.492	0.341	0.937	1.3E-5	5.3E-14
	DFM	0.156	0.360	0.947	0.573	5.7E-6	5.7E-6

Table 4: Directional Forecast Accuracy Statistics, Monthly Inflation

Notes: Case 1 is right before the start of the month. Case 2 is day 8 of the month. Case 3 is day 15 of the month, at which point the previous month's CPI is assumed to be available. Case 4 is day 22 of the month. Case 5 is the last day of the month, at which point the previous month's PCE price index is assumed to be available. Case 6 is day 15 of the following month, at which point the CPI for the month being nowcasted is assumed to be available. Success ratios report the percentage of nowcasts that correctly predicted whether inflation increased or decreased compared with the preceding monthly reading, and *p*-values come from the Pesaran-Timmermann test of directional forecast accuracy. **Bold** entries denote the highest success ratio for each inflation measure in each case. PCE and core PCE statistics exclude September and October 2001. The exercise uses real-time data from September 2000 through June 2015.

²⁰ Our combination of monthly data and real-time data limit the number of alternative models we pursue. For example, another approach could be to extract principal components from a dataset of the subcomponents of headline or core inflation and generate forecasts as in Stock and Watson (2002a, 2002b). But real-time databases such as ALFRED have extremely limited histories of monthly subcomponents of the PCE price index; the same is true for the seasonally adjusted subcomponents of the CPI.

While the results above use the entire sample of real-time data from September 2000 to June 2015, we ran a split-sample exercise as well, with the early sample running from September 2000 to December 2007 and the late sample running from January 2008 to June 2015. We find that the patterns of RMSEs were not markedly different between the early and the late sample, and our basic results still hold: our nowcasting model consistently outperforms the MIDAS and dynamic factor models for headline inflation and is comparable for core inflation.

Nowcasting Year-Over-Year Inflation

Modugno (2013) and Monteforte and Moretti (2013) both nowcast year-over-year inflation rates rather than monthly inflation rates. Using the same underlying monthly models and the same cases described above, Table 5 assesses the ability of the models to nowcast year-over-year inflation.²¹ Our same basic results hold, so we omit results on directional forecast accuracy: there continue to be only two cases in which the MIDAS models and the dynamic factor model produce smaller RMSEs than our baseline model, and the outperformance of our baseline model is notable for headline CPI and PCE inflation and smaller for core inflation measures.

In the closest related work to this paper, Modugno (2013) uses a dynamic factor model to nowcast year-over-year inflation in the headline CPI between January 2001 and December 2011, a slightly shorter sample than ours. During that sample, Modugno (2013) reports a nowcast RMSE of 0.23 percentage point on the day after the previous month's CPI is released, a 56.6 percent improvement over the 0.53 percentage point RMSEs from a random walk model in

²¹ Our random walk model now assumes a random walk in year-over-year inflation rather than in monthly inflation. Our baseline model, the MIDAS models, and the dynamic factor model continue to use monthly inflation rates, and then we compute the implied year-over-year inflation rates.

which year-over-year inflation is expected to remain unchanged from its previous reading. Using the same sample period and focusing on the day after the previous month's CPI is released, our model produces RMSEs of 0.16 percentage point for year-over-year CPI inflation, for a further 30.4 percent reduction in RMSE.

		Case						
Measure	Model	1	2	3	4	5	6	
CPI	Baseline	0.361	0.280	0.175	0.157	0.154		
	Random walk	0.830***	0.830***	0.497***	0.497***	0.497***		
	MIDAS	0.466***	0.476***	0.278***	0.279***	0.267***		
	DFM	0.460***	0.427***	0.277***	0.279***	0.287***		
Core CPI	Baseline	0.153	0.153	0.100	0.100	0.100		
	Random walk	0.193***	0.193***	0.124***	0.124***	0.124***		
	MIDAS	0.157	0.156	0.102	0.102	0.102		
	DFM	0.156	0.157	0.101	0.102	0.102		
PCE	Baseline	0.303	0.260	0.219	0.209	0.157	0.137	
	Random walk	0.595***	0.595***	0.595***	0.595***	0.364***	0.364***	
	MIDAS	0.383***	0.396***	0.300***	0.299***	0.214***	0.128*	
	DFM	0.357***	0.335***	0.336***	0.337***	0.226***	0.221***	
Core PCE	Baseline	0.215	0.215	0.196	0.196	0.144	0.129	
	Random walk	0.266**	0.266**	0.266***	0.266***	0.191**	0.191**	
	MIDAS	0.233	0.237	0.197	0.194	0.147	0.129	
	DFM	0.206	0.209	0.210*	0.211*	0.144	0.144***	

Table 5: Root Mean Squared Nowcast Errors, Year-Over-Year Inflation

Notes: Case 1 is right before the start of the month. Case 2 is day 8 of the month. Case 3 is day 15 of the month, at which point the previous month's CPI is assumed to be available. Case 4 is day 22 of the month. Case 5 is the last day of the month, at which point the previous month's PCE price index is assumed to be available. Case 6 is day 15 of the following month, at which point the CPI for the month being nowcasted is assumed to be available. *, **, and *** denote rejection of the null of equal conditional predictive ability for the baseline model compared with each alternative model at the 10%, 5%, and 1% level, respectively, based on the Giacomini-White test. **Bold** entries denote that the baseline model produces smaller RMSEs than the alternative model. Inflation rates are year-over-year percent changes, so errors are expressed in percentage points. PCE and core PCE statistics exclude September and October 2001. The exercise uses real-time data from September 2000 through June 2015.

Nowcasting Quarterly Inflation

Because there are a larger number of data releases over the course of a quarter, we

illustrate the baseline model's nowcasting performance for quarterly inflation at 14

representative dates, diagrammed in Figure 3.



Figure 3: Data Flow Timing

We compare the model with a number of alternatives. The three quarterly statistical models we consider have respectable inflation forecasting properties (see Faust and Wright 2013).²² All forecasts are made using the data that would have been available in real time; e.g., data release lags imply that the last available quarterly inflation reading at the very beginning of a quarter would actually be from two quarters earlier. Where necessary, the targeted quarter *T* is forecasted recursively.

1. A four-quarter random walk, where today's expected annualized quarterly inflation rate is the inflation rate over the last four available quarters; e.g., if the most recent available observation was in quarter *T*-1, $E_T \pi_T = 100(P_{T-1} / P_{T-1-4} - 1)$, similar to Atkeson and Ohanian

(2001).

²² We also considered a variety of other quarterly models which we do not show, including: a quarterly random walk model; AR(1) and AR(4) models estimated using real-time five-year rolling windows; AR(1) and AR(4) models estimated on the entire expanding real-time sample; direct rather than recursive AR(1) and AR(4) models; and AR(p) models estimated using five-year rolling windows, the entire expanding real-time sample, or inflation in "gap" form estimated using the entire expanding real-time sample, where the choice of p was based on the AIC in real time. These models were generally outperformed by the models shown.

- 2. An AR(1) model in inflation gaps, $x_T = \alpha_0 + \alpha_1 x_{T-1} + e_T$, with $x_T = \pi_T \pi_T^{LR}$ (see Kozicki and Tinsley 2001, Cogley et al. 2010, Clark 2011, and Faust and Wright 2013). Long-run inflation expectations within a quarter, π_T^{LR} , are measured by the Blue Chip consensus inflation expectation five-to-ten years ahead that would have been available in real time and are assumed to follow a random walk in the future. We estimate the coefficients on real-time expanding samples with the first gap observation in the second quarter of 1984.²³
- 3. The unobserved components model with stochastic volatility (UC-SV) from Stock and Watson (2007). For each inflation series, we begin the estimation in the first quarter of 1960 and use the real-time data that would have been available at the time.

We also consider the two competing mixed frequency nowcasting models set out above that can take advantage of high-frequency weekly and daily data, again using the real-time data that would have been available at the time.

- 4. MIDAS models, similar in spirit to Monteforte and Moretti (2013).
- 5. A dynamic factor model, based on Modugno (2013).

Figure 4 shows the quarterly nowcast RMSEs from the model and the competing quarterly forecasting models and nowcasting models. The quarterly forecasting models show few changes in forecast accuracy across the cases as time goes by; these changes occur when

²³ The Blue Chip consensus reports long-run forecasts of CPI inflation and GDP deflator inflation. As in Faust and Wright (2013), we assume long-run forecasts of PCE inflation (and core PCE inflation) are equal to those for the GDP deflator, and long-run forecasts for core CPI inflation are equal to those for headline CPI inflation. The long-run forecasts are typically released in March and October. Because March is late in the first quarter, we assume the March forecasts were only available in real-time as of the second quarter. Long-run CPI forecasts first appeared in March 1983, were not reported in October 1983, then reappeared on a continuous basis starting in March 1984.

new or revised CPI and PCE data are released.²⁴ Because the short sample makes the analysis sensitive to outliers, we exclude the fourth quarter of 2008 when computing the RMSEs.²⁵



Figure 4: Root Mean Squared Nowcast Errors, Quarterly Inflation

Notes: See Figure 3 for the timing of the 14 cases. Quarterly inflation rates are seasonally adjusted annualized rates, so numbers are expressed in annualized percentage points. All RMSE statistics exclude 2008Q4. PCE and core PCE statistics also exclude 2001Q3 and 2001Q4. The exercise uses real-time data from 2000Q4 through 2013Q4.

Figures 4(a) and 4(c) show that the baseline model's nowcasts for headline inflation-

whether measured by the CPI or the PCE price index-tend to broadly outperform the forecasts

²⁴ The largest revisions to the quarterly statistical forecasts' accuracy occur in case 3 for CPI and case 5 for PCE, when the third monthly reading for the previous quarter is released thus completing the quarter and the forecasts are conducted using an additional data point. Other changes in forecast accuracy reflect data revisions.

²⁵ Quarterly CPI inflation went from 6 percent at an annual rate in 2008Q3 to -9 percent in 2008Q4. The statistical models completely fail to predict this swing in inflation, with absolute errors near 15 percentage points. By contrast, this paper's nowcasting model quickly picks up the depths of the swing: by the middle of 2008Q4, the model was nowcasting headline CPI inflation of -7 percent, and the nowcast had fallen to -9 percent by the end of the quarter.

and nowcasts from a variety of models, both quarterly forecasting models and alternative mixedfrequency nowcasting models. The outperformance is apparent even immediately prior to the start of the quarter (case 1). During the first month of the quarter, the arrival of high-frequency readings on gasoline and oil prices helps to reduce the model's nowcast errors by about onethird. Nowcasting errors decrease as the quarter goes along and more information is accumulated, with a considerable improvement in CPI nowcasting accuracy once the first monthly CPI report of the quarter is released (case 7). Immediately prior to the release of the quarterly inflation rate, the typical error for headline CPI and PCE inflation is approximately ¹/₄ percentage point at an annual rate. The model quickly shows large improvements compared with the best competing quarterly forecasting models; by the end of the first month of the quarter being nowcasted (case 5), the model's RMSEs are about one-half those from the AR(1) gap model and the UC-SV model. Meanwhile, the model's improvements over the best alternative mixed-frequency nowcasting model are smaller, but the improvements are nevertheless persistent throughout the quarter.

The model uses a smaller number of variables for core inflation readings, and as a result changes in the core inflation nowcasts occur less frequently. Core CPI nowcasts depend only on the history of the series, so changes coincide with CPI releases (cases 3, 7, and 11). With each subsequent new CPI release, nowcasting accuracy improves, as shown in Figure 4(b). Meanwhile, core PCE inflation relies on a combination of past core PCE inflation and core CPI readings, if the latter have an additional month of data. Consequently, core PCE inflation nowcasting accuracy improves with each additional CPI or PCE release, as shown in Figure 4(d). As with headline inflation, immediately prior to the release of the quarterly inflation rate the typical error for core inflation is 0.1 to 0.3 percentage point at an annual rate. The model's

30

nowcasts for core inflation rates are considerably more accurate than those coming from quarterly forecasting models and are highly competitive with those from the MIDAS model and the dynamic factor model.²⁶

V. Nowcasting Horseraces with Professional Forecasters

Faust and Wright (2009, 2013) show that professional forecasters' inflation nowcasts tend to outperform those from statistical models.²⁷ In fact, Faust and Wright (2013) suggest that subjective nowcasts may hold a distinct advantage through their ability to "add expert judgment" to models (p. 20). Improved nowcasts are not only of interest for their own sake: Faust and Wright (2013) and Del Negro and Schorfheide (2013) show that taking advantage of more accurate inflation nowcasts improves inflation forecasting accuracy at longer horizons. Thus, we test our inflation nowcasting model by competing with other forecasters.

We compare the model's nowcasts with three benchmarks. The first two comparisons come from private forecasters that are available contemporaneously in real-time: the monthly Blue Chip Economic Indicators survey and the quarterly Survey of Professional Forecasters (SPF) compiled by the Philadelphia Fed. The final comparison uses inflation nowcasts from the Federal Reserve Board's Greenbook, which are released to the public with a 5-year delay.

 $^{^{26}}$ In the interest of space, we omit formal statistical results for all measures and all quarterly cases. For headline inflation, Giacomini-White tests reject the null of equal conditional predictive ability between the model and the alternative models at conventional significance levels in the vast majority of cases for both CPI and PCE inflation, and the model generates considerably higher success ratios and lower *p*-values based on the Pesaran-Timmermann directional forecast accuracy test. For core inflation measures, the model exhibits similar improvements over the quarterly forecasting alternatives, while Giacomini-White and Pesaran-Timmermann results are less conclusive compared with the alternative mixed-frequency models.

²⁷ Ang et al. (2007) examine forecasts of four-quarter inflation—which are importantly influenced by the nowcast and similarly find strong support for survey inflation forecasts over a number of model-based forecasts.

Across all the comparisons, we ensure identical information sets: we match the dates when the surveys or Greenbook forecasts were conducted with the real-time data available for the model's nowcasts. We show both RMSEs for the model's nowcasts and nowcasts from other forecasters to give a sense of absolute errors, along with ratios of mean-squared errors expressed in terms of the professional forecasters' errors relative to those from the model.

The model's nowcasts in many cases outperform professional forecasters. Real-time data availability limits the comparisons to a relatively short time span, with the earliest comparisons in 1999. Nevertheless, the model's nowcasting accuracy for headline inflation tends to easily outperform the Blue Chip consensus and the SPF median, especially the former as the quarter goes on, and the model's headline inflation nowcasting accuracy is comparable to the accuracy of the Greenbook. Meanwhile, core inflation nowcasting accuracy from the model is highly comparable to nowcasts made by private forecasters or the Board staff.

Comparison with the Blue Chip Economic Indicators Survey

The Blue Chip Economic Indicators survey of private professional forecasters provides forecasts of major U.S. economic indicators, including quarterly CPI inflation. Blue Chip forecasts start with the first quarter for which complete data are not yet available, which allows for nowcasting comparisons. Blue Chip consensus forecasts are averages. The Blue Chip survey is typically released around the 10th of each month, but the survey is conducted over an earlier two-day period that is usually mentioned in the release; we match this timing in our model.²⁸

²⁸ When the Blue Chip survey dates are not listed, we assume the survey date was the first Thursday of the month. If the first Thursday is the first day of the month, we assume the survey date was the first Tuesday of the month.

Given the timing of the Blue Chip survey and the publication of CPI data, we compare Blue Chip nowcasting accuracy with the model at four different points in time for each quarter. For example, nowcasts of the first quarter are collected in the January, February, March, and April Blue Chip surveys; the April Blue Chip survey date is about one to two weeks before the BLS releases all the data needed to compute first quarter CPI inflation.²⁹ The nowcast evaluation spans the second quarter of 1999 through the second quarter of 2015.

As information over the quarter accumulates and we move from Month 1 (at the very beginning of the quarter) through Month 4 (the survey from the month immediately following the quarter, right before the quarterly CPI is available), nowcasting accuracy improves for both the Blue Chip consensus and the nowcasting model. Table 6 shows monotonic reductions in RMSEs from both nowcasts across the four cases. However, the model's nowcasts are more accurate on average than Blue Chip nowcasts at each point in time, with somewhat lower RMSEs in months 1 and 4 and substantially lower RMSEs in months 2 and 3. In terms of directional forecast accuracy, the model produces higher success ratios and lower *p*-values from the Pesaran-Timmermann test for months 1, 2, and 3; by the fourth month of the quarter, both nowcasts always predict the sign of the change in inflation.

The time period under consideration contains a range of events, including the mild 2001 recession, a period of rising oil prices, the financial crisis and plunge in oil prices during a deep recession, and a moderate subsequent economic recovery. Faced with such events, judgmental nowcasts from professional forecasters may have had a large inherent advantage over model-based nowcasts, because the former could look outside the model and incorporate other information during rapidly changing circumstances. Given that a small number of variables—

²⁹ Compared with the quarterly exercise in the previous section and in Figure 3, the Blue Chip survey dates roughly correspond to cases 2, 6, 10, and 13.

six—determine the model's CPI nowcasts, this outperformance vis-à-vis Blue Chip is

particularly noteworthy.

Table 6: Blue Chip CPI Now	Weasting Comparisons Blue Chip survey conducted in Month 1 Month 2 Month 3 M 1.812 1.053 0.482 1 1.878 1.416 0.829 1 1.075 1.808 2.965 0.512 0.036 0.001 0.862 0.846 0.938 0.754 0.815 0.908				
	Blue	Chip surve	y conducte	d in:	
	Month 1	Month 2	Month 3	Month 4	
Model RMSE	1.812	1.053	0.482	0.270	
Blue Chip RMSE	1.878	1.416	0.829	0.399	
Ratio, average Blue Chip MSE to model MSE	1.075	1.808	2.965	2.190	
GW <i>p</i> -values for test of conditional predictive ability	0.512	0.036	0.001	0.003	
Model success ratio	0.862	0.846	0.938	1	
Blue Chip success ratio	0.754	0.815	0.908	1	
Model Pesaran-Timmermann <i>p</i> -values	4.9E-9	2.6E-8	3.2E-12	NA	
Blue Chip Pesaran-Timmermann <i>p</i> -values	3.8E-5	3.0E-7	1.1E-10	NA	

Notes: Comparisons are matched based on Blue Chip survey dates; e.g., when nowcasting the first quarter, month 1 would refer to the Blue Chip survey date in January, month 2 would be February's date, and month 3 would be March's date. The Blue Chip survey in month 4 (e.g., April) is conducted prior to the availability of CPI inflation data for the previous quarter and is the final nowcast. Quarterly inflation rates are seasonally adjusted annualized percent changes, so errors are expressed in annualized percentage points. GW denotes Giacomini-White. Success ratios report the percentage of nowcasts that correctly predicted whether inflation increased or decreased compared with the preceding quarterly reading, and *p*-values come from the Pesaran-Timmermann test of directional forecast accuracy. The exercise uses real-time data from 1999Q2 through 2015Q2.

Comparison with the Survey of Professional Forecasters

The SPF is published quarterly and is released around the middle of the second month of the quarter. The Federal Reserve Bank of Philadelphia publishes the survey dates; these dates are about one week prior to the release date, which means that SPF nowcasts of current quarter inflation are made before the first monthly CPI reading for the quarter is released.³⁰ We match information sets that would have been available to the professional forecasters with the model's information set. The SPF has a long history of reporting CPI forecasts, and we perform CPI nowcast comparisons beginning in the second quarter of 1999. The SPF started reporting core CPI inflation, headline PCE inflation, and core PCE inflation in the first quarter of 2007, and we conduct comparisons with these three series starting at that point. In all cases, we end the

³⁰ This roughly corresponds to Case 6 from the quarterly exercise in the previous section.

comparisons in the second quarter of 2015. We use the SPF median nowcasts to eliminate outliers and as a check on the Blue Chip consensus exercise, which uses averages.

Table 7 reports results. The model's nowcasts for headline CPI and PCE inflation outperform the SPF nowcasts by 0.40 percentage point and 0.28 percentage point on average, respectively, and the model is more successful along directional forecast accuracy metrics for headline inflation as well. Meanwhile, similar to some of the results presented above when comparing the model's core inflation nowcasts to those from MIDAS models and dynamic factor models, the nowcasting horseraces between the model and the professional forecast accuracy.

Table 7: Survey of Professional Forecasters Nowcasting Comparisons

	CPI	Core CPI	PCE	Core PCE		
Model RMSE	0.981	0.565	0.806	0.518		
SPF RMSE	1.381	0.577	1.089	0.504		
Ratio, average SPF MSE to model MSE	1.980	1.043	1.823	0.948		
GW <i>p</i> -values for test of conditional predictive ability	0.009	0.758	0.007	0.747		
Model success ratio	0.877	0.824	0.941	0.706		
SPF success ratio	0.831	0.824	0.794	0.676		
Model Pesaran-Timmermann <i>p</i> -values	1.8E-09	4.2E-03	4.6E-07	0.027		
SPF Pesaran-Timmermann <i>p</i> -values	9.2E-08	3.8E-04	3.2E-04	0.219		

Notes: Real-time comparisons are based on the SPF survey dates. SPF expectations for each quarter are the median value. Quarterly inflation rates are seasonally adjusted annualized percent changes, so errors are expressed in annualized percentage points. GW denotes Giacomini-White. Success ratios report the percentage of nowcasts that correctly predicted whether inflation increased or decreased compared with the preceding quarterly reading, and *p*-values come from the Pesaran-Timmermann test of directional forecast accuracy. The CPI exercise uses real-time data from 1999Q2 through 2015Q2. The core CPI, PCE, and core PCE exercises use real-time data from 2007Q1 (the first available SPF estimates) through 2015Q2.

We view the results for headline and core inflation as perhaps somewhat surprising. The model's outperformance for headline PCE inflation confirms the earlier findings from the Blue Chip exercise for a second inflation measure and offers further evidence that judgmental nowcasts can be improved upon. The model's core inflation nowcasts reflect extreme parsimony. Nevertheless, the core inflation nowcasts coming from the SPF are quite similar to those from the model, whether we are looking at the statistics in Table 7 or the actual nowcasts themselves, which we show in the Appendix. This finding raises the possibility that professional

forecasters are using a similar method for nowcasting core inflation, suggesting that our model is essentially capturing professional forecasters' near-term inflation expectations; or, alternatively, that a variant of Atkeson and Ohanian (2001) is still a difficult inflation forecasting benchmark to beat.

Comparisons with the Federal Reserve Board's Greenbook

We also compare our model's nowcasts with those made by the staff economists at the Federal Reserve Board of Governors in the so-called Greenbook. A commonly held view is that Greenbook nowcasts and short-term forecasts for inflation are the gold standard. For example, Romer and Romer (2000) show that Greenbook inflation forecasts prior to 1991 were superior to those of private forecasters. Subsequent studies by Sims (2002) and Faust and Wright (2007) documented that current-quarter Greenbook inflation forecasts are on average superior to a variety of forecasting approaches.³¹ Bernanke (2007) describes the range of models, indicators, expertise, and extensive judgment used to inform Board staff's near-term inflation forecasts.

Greenbook forecasts are produced in the week prior to each of the Federal Open Market Committee's (FOMC) regularly scheduled meetings. We match the model's real-time information set to the Greenbook date. The eight FOMC meetings each year have historically been spaced irregularly, with essentially two meetings per quarter. For the sake of our exercise, we classify Greenbook nowcasts based on whether they were made in the first or second half of the quarter as H1 and H2, respectively; the different information sets available early or late in the

³¹ This contrasts with Branch (2014), wherein SPF inflation nowcasts are a proxy for monetary policymakers' nowcasts. In that context, improved nowcasts could affect Taylor rule estimates.

quarter make these effectively two different exercises.³² All four inflation measures in the paper are available in Greenbook, and our nowcasting exercise starts in the second quarter of 1999 for headline and core CPI inflation and the third quarter of 2000 for headline and core PCE inflation. The exercise ends in the fourth quarter of 2009, which is the last publicly available Greenbook.

Table 8 presents the comparisons. Nowcast errors across all inflation metrics decline dramatically from the first half of the quarter to the second half as additional information accumulates. With the exception of core CPI inflation in the first half of the quarter, the model's nowcast RMSEs are slightly larger than Greenbook's, and the success ratios are slightly lower. However, the differences in nowcasting accuracy are not large quantitatively with the exception of headline CPI inflation in H1. Additionally, conventional test statistics would fail to reject equal predictive ability in any of the cases considered. We interpret the statistical evidence—and the nowcasts themselves, which we show in the Appendix—as suggesting that our model's nowcasting accuracy is basically comparable to the combined judgment, modeling expertise, and resources devoted to inflation nowcasting in Greenbook for the time period under consideration.

		\mathcal{O}	1					
	C	PI	Core	CPI	PC	CE	Core	PCE
	H1	H2	H1	H2	H1	H2	H1	H2
Model RMSE	1.321	0.364	0.503	0.268	1.004	0.398	0.590	0.422
Greenbook RMSE	1.090	0.322	0.528	0.257	0.859	0.349	0.554	0.400
Ratio, average Greenbook MSE to model MSE	0.680	0.782	1.102	0.918	0.732	0.768	0.881	0.902
GW <i>p</i> -values for test of conditional predictive ability	0.176	0.364	0.509	0.615	0.255	0.241	0.378	0.485
Model success ratio	0.814	1	0.767	0.837	0.816	0.921	0.737	0.868
Greenbook success ratio	0.907	1	0.721	0.884	0.895	0.921	0.763	0.947
Model Pesaran-Timmermann <i>p</i> -values	3.9E-5	NA	1.7E-3	9.2E-5	2.1E-4	5.5E-7	1.6E-2	1.6E-5
Greenbook Pesaran-Timmermann <i>p</i> -values	7.6E-8	NA	7.5E-2	4.3E-6	2.8E-6	8.8E-7	7.9E-3	8.1E-8

Table 8: Greenbook Nowcasting Comparisons

Notes: Real-time comparisons are based on the Greenbook forecast dates. Forecasts made on or before the 20th day of the middle month of the quarter are in H1, and forecasts made after the 20th day of the middle month of the quarter are in H2. Quarterly inflation rates are seasonally adjusted annualized percent changes, so errors are expressed in annualized percentage points. GW denotes Giacomini-White. Success ratios report the percentage of nowcasts that correctly predicted whether inflation increased or decreased compared with the preceding quarterly reading, and *p*-values come from the Pesaran-Timmermann test of directional forecast accuracy. The CPI and core CPI exercises use real-time data from 1999Q2 through 2009Q4. The PCE and core PCE exercises use real-time data from 2000Q3 through 2009Q4. PCE and core PCE statistics exclude 2001Q3 and 2001Q4.

 $^{^{32}}$ Because of the irregular timing of Greenbook, we place the cutoff for H1 as on or before the 20th day of the middle month of the quarter.

VI. Assessing Nowcasting Accuracy and Sensitivity

To illustrate key drivers of the model's nowcasting accuracy, we consider robustness to the model and its assumptions—some small (e.g., changing rolling window lengths used in estimation) and others large (e.g., dropping disaggregates)—and show their effect on quarterly RMSEs in Table 9. The model's nowcasting performance is highly robust to minor variations, but nowcasting performance deteriorates when gasoline inflation is excluded from the set of relevant disaggregates and when very long windows are used to estimate parameters.

We first vary the length of estimation windows. To capture potential time-variation—in relationships between CPI and PCE inflation measures, between disaggregate and aggregate measures, as well as in unmodeled inflation components that are subsumed in constant terms— we use rolling windows of τ =24 monthly observations to estimate equations (2), (7), and (8). Our nowcasts benefit from short windows: modestly expanding or contracting the window length has a trivial impact on nowcasting accuracy, but accuracy deteriorates as the window grows; e.g., with τ =120 months (line 3), quarterly RMSEs for headline inflation increase 15 to 38 percent on average, and Greenbook headline inflation nowcasts notably outperform our model. We use a longer window (τ_L =60 monthly observations) to estimate the two-stage regression giving the long-run relationship between oil and gasoline prices. Nowcasts are trivially affected as this window expands or contracts modestly, but using very long windows or expanding windows on all observations (line 6) causes headline inflation nowcasting RMSEs to deteriorate by 15 to 16 percent early in the quarter. We also consider variations in *J*, which governs the number of terms used in forecasting via recursive moving averages. Increasing *J* from the baseline of 12

38

months causes a bifurcation: RMSEs for core CPI inflation tend to rise, while RMSEs for core

PCE inflation are slightly lower.

					· · · ·			
	C	PI	Core	e CPI	P	CE	Core	PCE
Alternative model assumptions	H1	H2	H1	H2	H1	H2	H1	H2
1. τ =12 months	1.02	1.01	1	1	1.05	1.04	1.01	1.03
2. τ =36 months	0.99	1.00	1	1	1.01	0.99	1.00	1.00
3. τ =120 months	1.25	1.38	1	1	1.19	1.15	1.00	1.01
4. τ_L =48 months	1.00	1.00	1	1	1.00	1.00	1	1
5. τ_L =72 months	1.00	1.00	1	1	0.99	1.00	1	1
6. τ_L =entire expanding real-time sample	1.16	1.05	1	1	1.15	1.03	1	1
7. $J=6$ months	1.05	1.01	1.11	1.01	1.07	1.04	1.14	1.06
8. <i>J</i> =24 months	1.00	1.03	1.03	1.04	0.97	1.00	0.95	0.98
9. <i>J</i> =36 months	1.00	1.03	1.05	1.05	0.98	1.01	0.96	0.99
10. Use CRB foodstuffs to nowcast π_t^{Food}	1.00	1.02	1	1	0.99	1.01	1	1
11. Drop π_t^{Food}	1.01	1.02	1	1	1.01	1.01	1	1
12. Use WTI instead of Brent crude oil prices	1.03	1.01	1	1	1.02	1.00	1	1
13. Drop oil prices as a predictor of π_t^{Gasoline}	1.11	1.02	1	1	1.09	1.01	1	1
14. Drop π_t^{Gasoline}	1.67	1.95	1	1	1.54	1.48	1	1
15. Extend oil prices as a predictor of π_t^{Gasoline}	1.05	1.01	1	1	1.04	1.00	1	1
16. Drop bridging equations	1.03	1.01	1	1	1.05	1.06	1.05	1.09
17. Single models, no model switching	1.05	1.01	1	1	1.08	1.06	1.05	1.09
18. Use AR(1) instead of $J=12$ month moving avg.	1.02	1.01	1.01	1.03	0.98	0.98	0.95	0.98

Table 9: Relative Quarterly RMSEs from Alternative Assumptions

Notes: Relative quarterly RMSEs are defined as the alternative model RMSE divided by the baseline model RMSE, so numbers greater than 1 imply higher RMSEs from the alternative model assumption(s). H1 (H2) reports the average relative quarterly RMSEs for the first (second) half of the quarter, which includes cases 1 through 7 (cases 8 through 14) as defined in Figure 3. The baseline model features τ =24 months, τ_L =60 months, and *J*=12 months.

We also consider changes to the disaggregates. We first examine the role of food inflation in the model. The Commodity Research Bureau (CRB) produces a daily spot commodity price index for foodstuffs, which is potentially a source of high-frequency data that could be used to help nowcast π_t^{Food} . Line 10 shows that using CRB foodstuffs spot prices to help nowcast monthly CPI food inflation—instead of the moving average approach used to forecast CPI food inflation in the baseline model—has little effect on the accuracy of our headline inflation nowcasts.³³ Dropping π_t^{Food} as a disaggregate in nowcasting headline inflation

³³ In keeping with the parsimonious nature of our model, we considered various simple models relating CRB foodstuffs to CPI food inflation. Preliminary analysis showed a lag between seasonally adjusted foodstuffs spot prices and CPI food inflation. To exploit this lag, we bridge from monthly CRB foodstuffs spot price inflation in month t-j to CPI food inflation in month t, where we choose j to maximize R^2 at each point in time. We include CPI food inflation in month t-1 as an additional regressor. Alternative model specifications—including directly

also has a minor impact on RMSEs (line 11). We next examine the role of gasoline and oil prices in the model. Using WTI spot oil prices instead of Brent as our measure of crude oil prices has essentially no impact on our results (line 12), despite the fact that a wedge between the two measures opened up during our sample period. Excluding energy price measures has a large effect on the accuracy of our model. If daily oil prices are excluded from the model (line 13), RMSEs increase 9 to 11 percent in the first half of the quarter, suggesting that current oil prices help predict future gasoline prices and, by extension, their influence on inflation.³⁴ Dropping π_l^{Gasoline} as a disaggregate (line 14) causes a large deterioration in nowcast accuracy, as RMSEs for headline inflation increase 48 to 95 percent.

Finally, we consider changes to the model's structure. Assuming that the last observed oil prices are useful predictors of gasoline prices far into the future—compared with our baseline assumption that they are only useful for one month—allows for computing arbitrary $\hat{\pi}_{t+k}^{\text{Gasoline}}$ for $k \ge 0$ and eliminates equation (9); doing so causes a minor increase in RMSE (line 15). A second potential change to the model structure is to drop the bridging of core CPI to core PCE and headline CPI to headline PCE during the interim when the previous month's CPI is available but before PCE readings are released, thereby eliminating equations (2) and (7). Nowcasts of CPI and core CPI are unaffected, but RMSEs for PCE and core PCE rise modestly (line 16), suggesting that such a bridging approach assists with nowcasting. The third structural change combines the previous two by extending the use of oil prices and dropping the bridging equations (3) is the single model for core inflation nowcasts and equation (8) is

using foodstuffs inflation as the measure of food inflation in our model, error correction models relating foodstuffs to the CPI for food, and different assumptions for how we bridged foodstuffs to CPI food inflation—had little effect on the results.

³⁴ In this case, the two-stage regression relating oil and gasoline prices is omitted and $\hat{\pi}_t^{\text{Gasoline}}$ only enters the model if there are at least some weekly data on gasoline prices within month *t*.

the single model for headline inflation nowcasts (line 17). RMSEs are modestly higher than the baseline, consistent with gains for the types of deterministic model switching we propose. Finally, we consider an alternative in which we replace the default for making forecasts of monthly variables as recursive 12-month moving averages (and hence coefficients 1/12) in equations (3) and (9) with an AR(1) model estimated over a rolling window of length τ =24 months. Doing so has a very modest impact, slightly worsening our nowcasts of core and headline CPI but improving our nowcasts of core and headline PCE.³⁵

VII. Conclusion

This paper develops a new model for nowcasting U.S. headline and core inflation. The model is relatively parsimonious, relying on a small number of data series and simple univariate and multivariate regressions alongside time-varying weights on disaggregate and aggregate variables that take advantage of the state of the information flow over the course of a month or quarter. These features contrast with some other nowcasting approaches that utilize larger datasets to extract common factors. Similar to these other approaches, however, we show that nowcasts of both monthly and quarterly inflation improve as time passes and additional information arrives.

In head-to-head comparisons using real-time data, the model's nowcasts often outperform various statistical models and arguably the best available alternatives: nowcasts from professional forecasters. In particular, the model's nowcasts of headline CPI and PCE inflation generally are more accurate than those from either the Blue Chip consensus or the Survey of Professional Forecasters as well as competing MIDAS and dynamic factor models used for

³⁵ While not reported, using additional AR terms to forecast the disaggregates tends to worsen our quarterly nowcasts, especially for CPI inflation measures. Estimating an AR(1) model using a rolling window of length τ produces smaller nowcast RMSEs than if we instead used an AR(*p*) model, where *p* was chosen based on the AIC.

nowcasting, and they rival the nowcasting accuracy of the Greenbook. The accuracy of the model's nowcasts for core CPI and PCE inflation, which are made using very simple univariate and multivariate techniques, are essentially on a par with those from more sophisticated and computationally intensive MIDAS models and dynamic factor models and the expert judgment used in the SPF and the Greenbook.

Given the well-documented difficulties in forecasting inflation, the model developed in this paper has the potential to reduce both nowcasting errors and longer-horizon forecasting errors for academic economists and professional forecasters. An open question for further investigation is whether a similar model with time-varying weights on disaggregate and aggregate components could be useful for nowcasting other series, such as GDP. The approach that we follow has also stressed the principle of parsimony in nowcasting inflation, relying on very few data series. Bringing additional data to bear—for example, by relying on disaggregate information from core goods and core services in nowcasting core inflation, or drilling down to a fine level of disaggregation to assist in bridging from core CPI inflation to core PCE inflation prior to the release of the latter-has the potential to improve nowcasting accuracy even further as real-time data availability increases. In addition, it is worth noting that our nowcasting exercise takes place in an era of anchored long-run inflation expectations, which played a role in dampening the volatility of core inflation readings. Additional empirical work would be needed to examine the extent to which a framework similar to ours would generate good nowcasts during a period of unanchored long-run inflation expectations.

VIII. References

Alquist, Ron and Lutz Kilian (2010) "What Do We Learn from the Price of Crude Oil Futures?" *Journal of Applied Econometrics* 25(4): 539–73.

42

Andreou, Elena, Eric Ghysels, and Andros Kourtellos (2013) "Should Macroeconomic Forecasters Use Daily Financial Data and How?" *Journal of Business and Economic Statistics* 31(2): 240–51.

Ang, Andrew, Geert Bekaert, and Min Wei (2007) "Do Macro Variables, Asset Markets, or Surveys Forecast Inflation Better?" *Journal of Monetary Economics* 54(4): 1163–1212.

Atkeson, Andrew and Lee E. Ohanian (2001) "Are Phillips Curves Useful for Forecasting Inflation?" Federal Reserve Bank of Minneapolis *Quarterly Review* (Winter): 2–11.

Bańbura, Marta and Michele Modugno (2012) "Maximum Likelihood Estimation of Factor Models on Datasets with Arbitrary Pattern of Missing Data" *Journal of Applied Econometrics* 29(1): 133–60.

Bańbura, Marta, Domenico Giannone, Michele Modugno, and Lucrezia Reichlin (2013) "Nowcasting and the Real-Time Data Flow" in Graham Elliott and Allan Timmermann, eds., *Handbook of Economic Forecasting*, vol. 2, Elsevier: North Holland.

Baumeister, Christiane, Pierre Guérin, and Lutz Kilian (2015) "Do High-Frequency Financial Data Help Forecast Oil Prices? The MIDAS Touch at Work" *International Journal of Forecasting* 31(2): 238–52.

Bernanke, Ben S. (2007) "Inflation Expectations and Inflation Forecasting" speech at the Monetary Economics Workshop of the National Bureau of Economic Research Summer Institute, Cambridge, Massachusetts. Available online: http://www.federalreserve.gov/newsevents/speech/bernanke20070710a.htm

Branch, William A. (2014) "Nowcasting and the Taylor Rule" *Journal of Money, Credit and Banking* 46(5): 1035–55.

Carriero, Andrea, Todd E. Clark, and Massimiliano Marcellino (2015) "Real-Time Nowcasting with a Bayesian Mixed Frequency Model with Stochastic Volatility" *Journal of the Royal Statistical Society*, Series A, 178(4): 837–62.

Clark, Todd E. and Michael W. McCracken (2009) "Tests of Equal Predictive Ability With Real-Time Data" *Journal of Business and Economic Statistics* 27(4): 441–54.

Clark, Todd E. (2011) "Real-Time Density Forecasts from Bayesian Vector Autoregressions with Stochastic Volatility" *Journal of Business and Economic Statistics* 29(3): 327–41.

Cogley, Timothy, Giorgio E. Primiceri, and Thomas J. Sargent (2010) "Inflation-Gap Persistence in the U.S." *American Economic Journal: Macroeconomics* 2(1): 43–69.

Crone, Theodore M., N. Neil K. Khettry, Loretta J. Mester, and Jason A. Novak (2013) "Core Measures of Inflation as Predictors of Total Inflation" *Journal of Money, Credit and Banking* 45(2-3): 505–19.

Del Negro, Marco and Frank Schorfheide (2013) "DSGE Model-Based Forecasting" in Graham Elliott and Allan Timmermann, eds., *Handbook of Economic Forecasting*, vol. 2, Elsevier: North Holland.

Diebold, Francis X. and Roberto S. Mariano (1995) "Comparing Predictive Accuracy" *Journal* of Business and Economic Statistics 13(3): 253–63.

Faust, Jon and Jonathan H. Wright (2007) "Comparing Greenbook and Reduced Form Forecasts using a Large Realtime Dataset" NBER Working Paper number 13397.

Faust, Jon and Jonathan H. Wright (2013) "Forecasting Inflation" in Graham Elliott and Allan Timmermann, eds., *Handbook of Economic Forecasting*, vol. 2, Elsevier: North Holland.

Ghysels, Eric (2015) "MIDAS Matlab Toolbox" unpublished manuscript.

Ghysels, Eric, Pedro Santa-Clara, and Rossen Valkanov (2004) "The MIDAS Touch: Mixed Data Sampling Regression Models" unpublished working paper.

Ghysels, Eric, Pedro Santa-Clara, and Rossen Valkanov (2005) "There Is a Risk-Return Trade-Off After All" *Journal of Financial Economics* 76(3): 509–48.

Giannone, Domenico, Lucrezia Reichlin, and David H. Small (2006) "Nowcasting GDP and Inflation: The Real-Time Informational Content of Macroeconomic Data Releases" European Central Bank Working Paper Series No 633.

Giannone, Domenico, Lucrezia Reichlin, and David Small (2008) "Nowcasting: The Real Time Information al Content of Macroeconomic Data Releases" *Journal of Monetary Economics* 55(4): 665–76.

Harvey, David, Stephen Leybourne, and Paul Newbold (1997) "Testing the Equality of Prediction Mean Squared Errors" *International Journal of Forecasting* 13(2): 281–91.

Hendry, David F. and Kirstin Hubrich (2011) "Combining Disaggregate Forecasts or Combining Disaggregate Information to Forecast an Aggregate" *Journal of Business and Economic Statistics* 29(2): 216–227.

Kozicki, Sharon and P.A. Tinsley (2001) "Shifting Endpoints in the Term Structure of Interest Rates" *Journal of Monetary Economics* 47(3): 613–52.

Lütkepohl, Helmut (2010) "Forecasting Nonlinear Aggregates and Aggregates with Timevarying Weights" CESifo Working Paper No. 3031. Modugno, Michele (2013) "Now-casting Inflation Using High Frequency Data" *International Journal of Forecasting* 29(4): 664–75.

Monteforte, Libero and Gianluca Moretti (2013) "Real-Time Forecasts of Inflation: The Role of Financial Variables" *Journal of Forecasting* 32(1): 51–61.

Peach, Richard, Robert Rich, and M. Henry Linder (2013) "The Parts Are More Than the Whole: Separating Goods and Services to Predict Core Inflation" Federal Reserve Bank of New York *Current Issues in Economics and Finance* 19(7).

Pesaran, M. Hashem and Allan Timmermann (2009) "Testing Dependence Among Serially Correlated Multicategory Variables" *Journal of the American Statistical Association* 104(485): 325–37.

Romer, Christina D. and David H. Romer (2000) "Federal Reserve Information and the Behavior of Interest Rates" *American Economic Review* 90(3):429–57.

Sims, Christopher A. (2002) "The Role of Models and Probabilities in the Monetary Policy Process" *Brookings Papers on Economic Activity* 33(2): 1–62.

Stock, James H. and Mark W. Watson (2003) "Forecasting Output and Inflation: The Role of Asset Prices" *Journal of Economic Literature* 41(3): 788–829.

Stock, James H. and Mark W. Watson (2002a) "Forecasting Using Principal Components from a Large Number of Predictors" *Journal of the American Statistical Association* 97(460): 1167–79.

Stock, James H. and Mark W. Watson (2002b) "Macroeconomic Forecasting Using Diffusion Indexes" *Journal of Business and Economic Statistics* 20(2): 147–62.

Stock, James H. and Mark W. Watson (2007) "Why Has Inflation Become Harder to Forecast?" *Journal of Money, Credit and Banking* 39(S1): 3–33.

Tulip, Peter (2009) "Has the Economy Become More Predictable? Changes in Greenbook Forecast Accuracy" *Journal of Money, Credit and Banking* 41(6): 1217–31.

IX. Appendix: Model and Professional Forecasters' Nowcasts

This appendix shows the actual nowcasts coming from the model with those from professional forecasters, using matched information sets as in Section V.

Figure A1 plots the competing nowcasts from the model and Blue Chip for each of the four cases set out above along with the actual quarterly CPI inflation rate. The model's nowcasts are very effective in tracking actual CPI inflation in all four cases, especially as time goes on and more information is available. However, the outperformance of the model is not universal, as Blue Chip nowcasts were sometimes more accurate than those from the model.

Figure A2 plots the nowcasts from the model and the SPF along with the inflation data for each series. As with the Blue Chip comparison, the model does not uniformly beat the SPF nowcasts for headline inflation. Rather, the model's outperformance relates to its ability to capture the volatility in inflation; this is especially apparent in the shorter sample in Figure 7(c), where the SPF nowcasts tend to be too stable compared with realized inflation.

Figure A3 plots the nowcasts from the model and the Greenbook, separating nowcasts based on whether they were made in H1 or H2 of each quarter. By H2, the model's nowcasts are typically very close to the Greenbook across all inflation measures and across the entire sample.

Figure A4 provides a real-time nowcasting illustration using headline CPI inflation in the second quarter of 2013. In the beginning of the quarter in April, the Blue Chip consensus nowcast was 1.8 percent at an annual rate. The average of the ten highest forecasts was above 2 percent, and the average of the ten lowest forecasts was below 1 percent. By mid-May, the Blue Chip consensus nowcast was 1.5 percent; around that same time, the median forecast from the SPF was 1.6 percent. The Blue Chip consensus nowcast fell to 0.5 percent by early June and to 0

percent in early July. We trace out the daily headline CPI nowcasts from our model and show that after starting the quarter near zero then falling off in mid-April, it began to move back toward 0 percent in the second half of the month. From late April through the end of the quarter, the model expected headline CPI inflation would be in the range of 0 to -0.5 percent at an annual rate. When the BLS released the June CPI report on July 16, headline CPI inflation for 2013Q2 came in just below 0 percent. In terms of absolute errors, the model outperformed the SPF and the Blue Chip consensus in three of four cases during the quarter.

Figure A1: Model and Blue Chip CPI Inflation Nowcasts



Notes: Cases are defined by the Blue Chip survey dates. Month 1 is the Blue Chip survey date for the first month of the quarter being nowcasted (e.g., January when nowcasting Q1). Month 2 is the Blue Chip survey date for the second month of the quarter being nowcasted. Month 3 is the Blue Chip survey date for the third month of the quarter being nowcasted. Month 4 is the Blue Chip survey date for the first month of the quarter following the quarter being nowcasted (e.g., April when nowcasting Q1). Quarterly inflation rates are seasonally adjusted annualized percent changes. The exercise uses real-time data from 1999Q2 through 2015Q2.



Notes: Real-time comparisons are based on the SPF survey dates. SPF expectations for each quarter are the median value. Quarterly inflation rates are seasonally adjusted annualized percent changes. The CPI exercise uses real-time data from 1999Q2 through 2015Q2. The core CPI, PCE, and core PCE exercises use real-time data from 2007Q1 (the first available SPF estimates) through 2015Q2.



Figure A3: Model and Greenbook Inflation Nowcasts



Figure A3 (continued): Model and Greenbook Inflation Nowcasts (g) Core PCE, first half of quarter (h) Core PCE, second half of quarter

Notes: Real-time comparisons are based on Greenbook forecast dates. Forecasts made on or before the 20th day of the middle month of the quarter are in H1, and forecasts made after the 20th day of the middle month of the quarter are in H2. Quarterly inflation rates are seasonally adjusted annualized percent changes, so numbers are expressed in annualized percentage points. The CPI and core CPI exercises use real-time data from 1999Q2 through 2009Q4. The PCE and core PCE exercises use real-time data from 2000Q3 through 2009Q4.



Figure A4: Real-Time Nowcasts of Headline CPI Inflation in 2013Q2

Notes: The Blue Chip marks show 2013Q2 nowcasts of headline inflation from the Blue Chip Economic Indicators surveys that were released in April, May, June, and July of 2013. The SPF median is for 2013Q2. The red line shows daily nowcasts of headline CPI inflation from the model. The solid black circle is the actual annualized CPI inflation rate reported by the BLS for 2013Q2.